Author's response to reviews

Title: Effectiveness of Lifestyle Change plus Dental Care (LCDC) program on improving glycemic and periodontal status in the elderly with type 2 diabetes

Authors:

Saruta Saengtipbovorn DDS MPH (m_mimmy602@hotmail.com)
Surasak Taneepanichskul MD (surasak.t@chula.ac.th)

Version: 5
Date: 4 June 2014

Author's response to reviews: see over
Dear Editorial board of BMC Oral Health,

Please find attached our revised manuscript “Effectiveness of Lifestyle Change plus Dental Care (LCDC) program on improving glycemic and periodontal status in the elderly with type 2 diabetes” (MS: 2072947886122603). All authors read and approved the revised manuscript. We appreciate the opportunity to revise our manuscript; thereby the comments of the reviewers were very helpful in order to improve the clarity of the paper. Our responses to the comments are given below. We hope that this revised version can contribute to the contents of BMC Oral Health and be of great interest to the journal’s international readership.

Kind regards
Saruta Saengtipbovorn DDS MPH
Dentist, Health center 54
Editor's comment:

1. In your ethics statement we note that ethics approval was sought; please amend your statement to indicate whether ethics approval was granted.

   The statement to indicate whether ethics approval was granted has been stated in the ethical consideration section with yellow highlight.

2. As your study also used an intervention please provide a trial registration number.

   The trial registration number has been provided in the abstract with yellow highlight.